Author’s response to reviews

Title: Description of the surgical technique for condylectomy with minimally invasive surgery to treat interdigital helomas on the lesser toes: a Delphi study

Authors:
Luis Miguel Marti (lmarti@umh.es)
Alba Gracia-Sánchez (albagpodo@gmail.com)
Javier Ferrer-Torregrosa (javier.ferrer@ucv.es)
Ruben Lorca-Gutierrez (dr.lorca@hotmail.es)
Jonatan Garcia-Campos (jgarcia@umh.es)
Salvador Sanchez-Perez (salva@umh.es)

Version: 2 Date: 05 Feb 2019

Author’s response to reviews:

Dear editor,

Thank you very much for the acceptance of our manuscript.

Please, find attached a clean version of the manuscript.

Regards,

Dr. Luis M Marti-Martinez